Author’s response to reviews

Title: Three years of pulmonary rehabilitation: inhibit the decline in airflow obstruction, improve exercise endurance time, and body-mass index, in chronic obstructive pulmonary disease

Authors:

David Stav (dstav@asaf.health.gov.il)
Meir Raz (raz_m@maccabi.org.il)
Isaac Shpirer (i_shpirer@yahoo.com)

Version: 9 Date: 27 May 2009

Author's response to reviews:

Dear editor

Table 2. is missing in the manuscript

david stsv